Author's response to reviews

Title: The ratio of means method as an alternative to mean differences for analyzing continuous outcome variables in meta-analysis: a simulation study.

Authors:

Jan O Friedrich (j.friedrich@utoronto.ca)
Neill KJ Adhikari (neill.adhikari@utoronto.ca)
Joseph Beyene (joseph@utstat.toronto.edu)

Version: 4 Date: 12 May 2008

Author's response to reviews:

May 12, 2008
Editorial Office
BMC Medical Research Methodology
Dear Sir/Madam:

RE: MS: 1233987214186979
The ratio of means method as an alternative to mean differences for analyzing continuous outcome variables in meta-analysis: a simulation study.
Jan O Friedrich, Neill KJ Adhikari and Joseph Beyene

We are delighted that BMC Medical Research Methodology will publish our manuscript. We are resubmitting a new version of our paper that includes all your requested formatting changes. If you need any additional information, please do not hesitate to contact us.

Sincerely,

Jan Friedrich, MD, DPhil
Critical Care and Medicine Departments,
St. Michael’s Hospital and University of Toronto
30 Bond Street, Bond Wing, Room 4-015
Toronto, Ontario
CANADA M5B 1W8
Tel: 416-864-5559
Fax: 416-864-6013
E-mail: j.friedrich@utoronto.ca